

SPECIFIC COMMENTS TO AUTHORS

The main manifestations of chronic heart failure are dyspnea, fatigue and fluid retention, which are the end-stage manifestations of cardiovascular disease and the main cause of death. In this study, the authors summarized the effective active ingredients of the Chinese herbal in the treatment of chronic heart failure. The mechanism of the Chinese herbal compound for invigorating the spleen, removing phlegm and removing blood stasis in the treatment of chronic heart failure was also explored. The study is overall well designed and performed. The results are very interesting. After a minor editing, this study can be accepted for publication. Comments:

1. The title seems too long, please short it.

Answers: The title has been revised accordingly.

2. Tables should be edited.

Answers: We edited the tables carefully.

3. Please take attention about some abbreviations.

Answers: All abbreviations are double checked.